

Micro-evolutionary response of spring migration timing in a wild seabird

Maria Moiron, Céline Teplitsky, Birgen Haest, A. Charmantier, Sandra

Bouwhuis

► To cite this version:

Maria Moiron, Céline Teplitsky, Birgen Haest, A. Charmantier, Sandra Bouwhuis. Micro-evolutionary response of spring migration timing in a wild seabird. Evolution Letters, In press, 10.1093/evlett/qrad014. hal-04248730

HAL Id: hal-04248730 https://hal.science/hal-04248730

Submitted on 18 Oct 2023

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

Micro-evolutionary response of spring migration timing in a wild seabird

Maria Moiron^{1,2,}, Céline Teplitsky², Birgen Haest³, Anne Charmantier², Sandra Bouwhuis^{1,1}

¹Life-history Biology Department, Institute of Avian Research, Wilhelmshaven, Germany ²CEFE, Univ Montpellier, CNRS, EPHE, IRD, Montpellier, France ³Department of Bird Migration, Swiss Ornithological Institute, Sempach, Switzerland

Corresponding author: Institute of Avian Research, An der Vogelwarte 21, 26386 Wilhelmshaven, Germany. Email: mariamoironc@gmail.com A.C. and S.B. contributed equally as last authors.

Abstract

In the context of rapid climate change, phenological advance is a key adaptation for which evidence is accumulating across taxa. Among vertebrates, phenotypic plasticity is known to underlie most of this phenological change, while evidence for micro-evolution is very limited and challenging to obtain. In this study, we quantified phenotypic and genetic trends in timing of spring migration using 8,032 dates of arrival at the breeding grounds obtained from observations on 1,715 individual common terns (Sterna hirundo) monitored across 27 years, and tested whether these trends were consistent with predictions of a micro-evolutionary response to selection. We observed a strong phenotypic advance of 9.3 days in arrival date, of which c. 5% was accounted for by an advance in breeding values. The Breeder's equation and Robertson's Secondary Theorem of Selection predicted qualitatively similar evolutionary responses to selection, and these theoretical predictions were largely consistent with our estimated genetic pattern. Overall, our study provides rare evidence for micro-evolution underlying (part of) an adaptive response to climate change in the wild, and illustrates how a combination of adaptive micro-evolution and phenotypic plasticity facilitated a shift towards earlier spring migration in this free-living population of common terns.

Keywords: adaptation, Breeder's equation, climate change, common tern, phenology, Robertson's Secondary Theorem of Selection

Lay Summary

Empirical evidence for evolutionary change underlying vertebrate adaptation to current global change is very rare. This may be due to phenotypic plasticity being the main mechanism underlying adaptation, or to challenges associated with the empirical testing of genetic changes in the wild, in particular data limitations. In this study, we tested whether an observed phenotypic advance in the timing of arrival from spring migration in a wild seabird population was due to an evolutionary response (i.e., genetic change) and/ or to phenotypic plasticity or other non-genetic effects. To do so, we applied a bivariate "animal model" to a 27-year data set from a pedigreed population of common terns located at the North Sea coast of Germany. We found an evolutionary response to selection favoring earlier arriving individuals. Additionally, we could show that two different theoretical models predict a qualitatively similar evolutionary response as the one we estimated, both in terms of direction and magnitude. As such, our study provides a rare empirical case where estimated and predicted evolutionary responses agree, and suggests an evolutionary response in the timing of avian spring migration, although it played a smaller role than phenotypic plasticity in the common tern response to rapid climate change. Overall, these findings show the use of disentangling the relative, and often complementary, contributions of plastic and evolutionary changes to better understand adaptive processes and predict responses to future changes.

Introduction

Phenotypic changes in response to climate change are commonplace across taxa (Bellard et al., 2012; Chen et al., 2011; Hendry et al., 2008; Parmesan, 2006; Parmesan & Yohe, 2003; Radchuk et al., 2019), but the mechanism(s) underlying these changes remain largely unknown (Gienapp & Brommer, 2014; Hoffmann & Sgrò, 2011; Lavergne et al., 2010; Merilä & Hendry, 2014; Salamin et al., 2010). Where the mechanisms underlying the observed shifts have been explicitly investigated, in vertebrates they have often been attributed to phenotypic plasticity (i.e., genotypes express variable phenotypes under different environmental conditions), while evidence for adaptive micro-evolution (i.e., genotypes that have a higher fitness increase in frequency in the population) is still relatively rare (Gienapp & Brommer, 2014; Gienapp et al., 2008; Hairston et al., 2005; Henderson, 1963; Hoffmann & Sgrò, 2011; Merilä & Hendry, 2014; Vedder et al., 2013; Visser, 2008).

Animal models (sensu Kruuk, 2004) are useful statistical tools to test for genetic trends and rates of micro-evolutionary responses to selection by means of providing estimates of individual breeding values (i.e., the expected effect of the genes that an individual passes on to its offspring, Falconer & Mackay, 1996; Lynch & Walsh, 1998). The strength of this test, however, relies on properly accounting for uncertainty in breeding value predictions and environmental factors (Hadfield et al., 2010; Postma, 2006).

Received February 14, 2022; revisions received February 6, 2023; accepted April 17, 2023

[©] The Author(s) 2023. Published by Oxford University Press on behalf of The Society for the Study of Evolution (SSE) and European Society for Evolutionary Biology (ESEN).

This is an Open Access article distributed under the terms of the Creative Commons Attribution License (https://creativecommons.org/licenses/by/4.0/), which permits unrestricted reuse, distribution, and reproduction in any medium, provided the original work is properly cited.

To our knowledge, there are only three long-term studies from natural vertebrate populations that have found evidence for a genetic change underlying a phenotypic change (most likely) in response to climate change, while taking into account the uncertainty in the breeding value estimation and controlling for a temporal trend in the phenotypic data (but see Merilä et al., 2001 and Walsh & Lynch, 2018 for reviews on evolutionary stasis, where no micro-evolutionary change was observed). First, based on 10 years of data on an alpine population of snow voles (Chionomys nivalis), Bonnet et al. (2017) showed a genetic change towards a reduced body mass as an adaptive response to viability selection, likely in response to a change in snowfall patterns. This genetic change represented a case of "cryptic evolution" where the population did evolve but there was no observed phenotypic change in body mass. Second, Evans & Gustafsson (2017) studied male plumage coloration in collared flycatchers (Ficedula albicollis) over 34 years and showed a marked phenotypic decline, which was mirrored by a decline in the mean breeding value of these males. This decline, which accounted for 11% of the total phenotypic change, was likely driven by viability selection acting on ornamentation being sensitive to the climatic conditions experienced at the breeding ground in the preceding year. And third, using data obtained from a wild population of red deer (Cervus elaphus), Bonnet et al. (2019) found that average parturition date had advanced over the 45-year study period, with warmer temperatures during the previous rut season associated with earlier parturition dates, and showed that a micro-evolutionary response accounted for 15% of the total advance of 12.3 days (Bonnet et al., 2019).

Once genetic trends have been estimated, testing their consistency with predictions of a micro-evolutionary response to selection can inform us about the occurrence and magnitude of adaptive responses to selection, as well as our ability to model these responses. Traditionally, the per-generation evolutionary response to selection in a given trait, R, can be predicted by $R = h^2S$, also known as the Breeder's equation, where h^2 represents the trait heritability, and S is the directional selection differential (Falconer & Mackay, 1996; Lynch & Walsh, 1998). This Breeder's equation, however, tends to lead to biased predictions in studies from natural populations due to the occurrence of correlated selection as well as confounding environmental effects (i.e., when selection is acting on the environmental rather than genetic component of a trait, the Breeder's equation will overestimate the trait's response to selection) (Morrissey et al., 2010). It therefore has been argued that micro-evolutionary responses to selection in natural populations are better predicted by applying Robertson's Secondary Theorem of natural Selection (hereafter, STS) (Morrissey et al., 2010), which postulates that the additive genetic covariance between a trait and relative fitness is a direct measure of the expected per generation evolutionary change in the mean trait value, R (Price, 1970; Robertson, 1966). While STS is a strictly genetic approach to predicting a response to selection, and therefore unbiased by unmeasured covariates, it is also agnostic regarding the selection processes underlying the micro-evolutionary responses. Hence, a comprehensive understanding of the predicted evolutionary responses to selection will benefit from evaluating the predictions from both the Breeder's equation and the Robertson's Secondary Theorem of Selection, as they provide complementary information.

Along with poleward shifts in geographic distributions, phenotypic changes in response to global warming are most apparent in phenology (Menzel et al., 2006; Parmesan, 2006; Poloczanska et al., 2013; Root et al., 2003; Thackeray et al., 2010). As such, timing of migration, together with timing of breeding, is a trait of special interest in the context of current global and climate change (Charmantier & Gienapp, 2014; Gienapp et al., 2014; Parmesan & Yohe, 2003). Among birds, most studied populations have shifted their phenology to earlier migration and breeding (review: Charmantier & Gienapp, 2014; Gienapp et al., 2007; Gordo, 2007; Lehikoinen et al., 2004; meta-analysis: Bailey et al., 2022; Radchuk et al., 2019), although not all bird species share the same adaptive potential (e.g., Keogan et al., 2018). For instance, long-distance migrant birds might be constrained in their potential to keep track of changing environmental conditions by inherited circannual clocks and migratory behaviors (Åkesson et al., 2017) and by reduced correlation between climate at the wintering and the breeding grounds, making them more sensitive to rapid warming conditions. With respect to the timing of migration, the date of arrival at the breeding grounds has received most empirical attention, and early arrival is expected to contribute to securing high-quality breeding sites (Kokko et al., 2004) and mates (Coppack & Both, 2002; Ludwigs & Becker, 2005), and to favor an early initiation of breeding (Moiron et al., 2020). Besides often being under strong directional selection (for being early), arrival date has also been found to harbor an important genetic component. For instance, arrival date heritabilities ranged from 0.11 to 0.32 among barn swallows (Hirundo rustica, Teplitsky et al., 2011), common terns (Sterna hirundo, Amaud et al., 2013), and great reed warblers (Acrocephalus arundinaceus, Tarka et al., 2015), with collective evidence across four bird species revealing a mean heritability of 0.43 (Pulido, 2007).

Even though the prerequisites for a micro-evolutionary response are often met (non-zero additive genetic variance and selection), to date we know of only two studies providing "indirect" evidence for a micro-evolutionary response of the timing of migration in natural bird populations. First, Teplitsky et al. (2011) used two long-term datasets on barn swallows from Spain and Denmark to show that, for the Spanish but not the Danish population, the Breeder's equation and Robertson's Secondary Theorem of Selection predicted qualitatively similar responses to selection (towards earlier arrival). It was, however, not estimated whether the study population had undergone substantial phenotypic or genetic change across years, precluding a comparison between observed and predicted responses to selection. Second, Helm et al. (2019) replicated an experimental study of the annual cycle of a long-distance migratory species, the pied flycatcher (Ficedula hypoleuca), after 21 years of warming and showed that migration timing had advanced by 9 days. This advance was also observed in a nearby natural population of flycatchers, such that, altogether, their results supported a role of micro-evolution in the earlier spring migration timing.

In our study, we applied a bivariate animal model to test for a micro-evolutionary response in the timing of spring arrival in a long-distance migratory bird. We used 27 years of data from a free-living, pedigreed population of common terns (Sterna hirundo). We had two main objectives: (a) to quantify whether there was a detectable genetic change in arrival date (i.e., a temporal trend in individual breeding values) and (b) to test whether this trend was consistent with predictions of micro-evolutionary responses to selection based on both the Breeder's equation and Robertson's Secondary Theorem of natural Selection. Previous studies from the same study population found that arrival date was heritable (Arnaud et al., 2013; Moiron et al., 2020) and under directional selection, with earlier arrival being associated with improved reproductive success (Arnaud et al., 2013, but see Ezard et al., 2007) and a higher probability of survival (Zhang et al., 2015b). Hence, although we acknowledge that evolutionary stasis often occurs, even when the prerequisites for micro-evolution are met (Merilä et al., 2001; Walsh & Lynch, 2018), we expected a micro-evolutionary response towards earlier arrival.

Methods

Study system and data collection

The data were collected as part of a long-term study of a common tern population located at the Banter See on the German North Sea coast (53° 30′ 40″ N, 08° 06′ 20″ E). The colony consists of six concrete islands, each surrounded by a 60-cm wall. An individual-based study was initiated in 1992, when 101 adult birds were caught and marked with individually numbered subcutaneously injected transponders. Since 1992, all locally hatched birds have similarly been marked with a transponder shortly prior to fledging and the presence and reproductive performance of marked individuals has been monitored with the help of antennae and following a standard protocol (Becker & Wendeln, 1997). Thanks to the automatic antenna system, affixed to the walls of the colony site since 1994, it is possible to record the arrival date from the wintering grounds of every marked bird in the population. From this, and as confirmed by a tracking study, it is known that the common terns of this population show highly repeatable migratory behavior (Kürten et al., 2022).

As part of the standard protocol, breeding birds are identified using portable antennae placed around each nest for 1–2 days during incubation, which is shared by both partners. Pairs can rear up to three chicks per brood and produce replacement clutches after loss of eggs or chicks. True second clutches (i.e., non-replacement clutches) are extremely rare (1.4% across the study period). Chicks are ringed at hatching and checked every 2–3 days throughout the breeding season until they fledge (at about 26 days; Becker & Wink, 2003) or perish.

Data selection

The phenotypic and fitness data used in this study were collected between 1994 and 2020. We focused on arrival date at the breeding area from the wintering grounds in West Africa (Becker et al., 2016; Kürten et al., 2022), a phenological trait that captures variation in the timing of spring migration and that is defined as the day of first return to the breeding grounds (1 January = 1) (Zhang et al., 2015a). Because no breeding can occur before a bird has arrived at the colony and recovered from migration, we removed observations of arrival date obtained from the antenna system that were not at least 10 days earlier than the egg laying date (n = 696 observations from 390 birds) as they are deemed faulty (Zhang et al., 2015a). The resulting dataset of arrival date included 8,032 observations from 1,715 individuals of known sex and age (mean number of observations per individual = 4.68, range = 1–21). The overall mean (\pm SD) arrival date across the 27 years was 118 ± 12.63 (~28 April).

We used adult lifetime reproductive success (LRS) as our fitness measure, and quantified it as the total number of fledglings locally produced in the lifetime of an adult individual (Supplementary Figure S1). Although we cannot directly observe an individual's death, we can reliably assume it, because breeders at the Banter See are highly site-faithful, as evidenced by the resighting probability of breeding individuals being close to one (Szostek & Becker, 2012), and 96% of breeders not skipping recording by the antenna system for two or more consecutive years after first reproduction (Bouwhuis et al., 2015; Zhang et al., 2015a). In addition to using LRS data from individuals assumed to be dead (i.e., not observed in 2019 and 2020; n = 946), we also included LRS data from those individuals that were still alive (i.e., observed in 2019 and/or 2020) but older than 10 years (n = 121)to avoid "cohort truncation" (i.e., to avoid excluding individuals "not at random" with respect to fitness, Hadfield, 2008; Morrissey et al., 2012a). We chose this threshold because the cumulative reproductive success of known dead birds at ages older than 10 and their lifetime reproductive success are known to be highly correlated (r > 0.8, Moiron et al., 2022). This age threshold also matches the biology of the population, as the mean lifespan is ~10 years (Szostek & Becker, 2012). LRS was, however, assigned as "missing" for birds that were observed in 2019 and/or 2020 and younger than 10 years (n = 648). As such, the dataset of adult LRS consisted of 1,067 observations from individuals of known age and sex. Details of the social pedigree used in the study can be found in the Supplementary Material.

Statistical analyses

To test for a trend in phenology across the study period, we ran a linear model with the annual mean arrival dates observed between 1994 and 2020 as a response variable, year as a continuous fixed effect, and assuming a Gaussian error distribution.

To test for a micro-evolutionary change in arrival date, we fitted a bivariate animal model with arrival date and LRS as response variables. Arrival date was modeled assuming a Gaussian error distribution, while LRS was modeled assuming an overdispersed Poisson error distribution with log-link function (Supplementary Figure S1). Because our fitness measure follows a log-normal distribution, variance estimates for absolute fitness on the latent scale data are equivalent to variance estimates directly on the data scale for relative fitness (Bonnet et al., 2019; Morrissey & Bonnet, 2019; de Villemereuil et al., 2016).

The bivariate animal model included random intercepts for individual identity linked to the relatedness matrix (V,), for maternal environmental effects (V_{MOTHER}; although maternal identity is only known for 38.78% of the 1,715 individuals, see Supplementary Material), and for among-cohort variation (V_{COHORT}; note that year of hatching is known for all individuals in this dataset). We also fitted individual identity not linked to the pedigree to account for repeated measures and estimate the permanent environmental effect (V_{PE}) , and year of breeding (V_{VEAP}) as random effects associated only with arrival date. Because, unlike arrival date, LRS has a single measure per individual, it is not possible to fit a covariance at all levels of variation (see Bonnet et al., 2019; Morrissey et al., 2012a). As such, we only modeled the additive genetic covariance between LRS and arrival date (COV,), the maternal environmental covariance between LRS and arrival date ($\text{COV}_{\text{MOTHER}}$), the cohort covariance between LRS and arrival date (COV_{COHORT}), and the covariance between the permanent environmental effects on arrival date and the residuals of LRS (as both represent covariance among individuals).

We fitted sex (categorical variable) as a fixed effect associated with both LRS and arrival date. As fixed effects associated only with arrival date, we fitted the linear and quadratic effects of age (continuous variable measured in years, Ezard et al., 2007; Zhang et al., 2015a), the linear effect of year of breeding (continuous variable, to control for the linear temporal trend in phenology), and sea surface temperature at the main wintering area (Kürten et al., 2022) (continuous variable) both as a main effect and in interaction with the linear and quadratic effects of age. We used data of sea surface temperature between 23 June and 27 July the year prior to arrival at the coast of Guinea and Sierra Leone, determined based on the approach developed by Haest et al. (2018, 2019, 2020). All continuous fixed effects (except age) were mean centered and variance standardized.

We estimated the heritability (h^2) of arrival date conditional to the variance explained by fixed effects as the proportion of the total phenotypic variance in arrival date explained by the additive genetic variance.

Estimating micro-evolutionary change and genetic drift

From the bivariate animal model described above, we extracted the best linear unbiased predictors (BLUPs) of breeding values of arrival date for each individual and tested for a temporal change. Temporal trends were assessed in two ways, by either regressing the breeding values of arrival date over an individual's mean breeding year or cohort (i.e., the year of hatching). We used recruited individual data so that results are comparable across the two analyses.

To calculate the probability that the observed change in breeding values differed from a scenario resulting from genetic drift, we simulated random breeding values for arrival date as per Hadfield et al. (2010), and fitted a linear regression to these random breeding values to obtain the temporal slopes due to drift for each posterior sample (temporal slopes were estimated both using an individual's mean breeding year and cohort). Because the distribution of temporal slopes due to random drift is expected to be zero-centered (sampling was neutral), we could then calculate the proportion of the posterior distribution of these drift slopes that returned a regression value more negative than the posterior mode of the observed temporal slopes.

Finally, the annual estimate of a micro-evolutionary response estimated as the change in breeding values was converted to a per-generation rate by multiplying it by the population's generation time of ~7.95 years. We estimated this generation time for common terns in our dataset as the mean age of parents of nestlings that later recruit in the population (Charlesworth, 1994), 8 years being in line with the generation time previously estimated in this species (Nisbet et al., 2020). We also converted the annual estimate of micro-evolution to units of standard deviation per generation by multiplying by generation time and dividing by phenotypic standard deviations. This way, the unit for evolutionary change was equivalent to a change measured in Haldanes (Hendry & Kinnison, 1999).

Predicting micro-evolutionary change

We predicted the expected per-generation rate of micro-evolutionary change of arrival date by first applying the Breeder's Equation (i.e., $R = h^2S$). The selection differential or phenotypic covariance between focal trait and fitness, S, was calculated as the sum of the additive genetic and permanent environmental covariances between arrival date and adult LRS. To obtain the full posterior distribution for the evolutionary response to selection (R) and hence account for the uncertainty in all estimated parameters, we multiplied the full posterior distributions of h^2 and S. Second, Robertson's Secondary Theorem of natural Selection states that the additive genetic covariance between a trait and relative fitness represents a direct measure of the expected per-generation evolutionary change R in that trait (Price, 1970; Robertson, 1966). Both estimates of evolutionary response to selection (R) were reported as the posterior mode and 95% credible intervals.

Statistical model implementation

We fitted the model using a Bayesian framework implemented in the statistical software R (v. 4.0.5, R Core Team, 2019) using the R-package MCMCglmm (Hadfield, 2010). We fitted a parameter-expanded prior. The number of iterations and thinning interval were chosen to ensure that the minimum MCMC effective sample size for all parameters was 1,000. Burn-in was set to a minimum of 2,000 iterations. The retained effective sample size yielded absolute autocorrelation values lower than 0.1 and satisfied convergence criteria based on the Heidelberger and Welch convergence diagnostic (Heidelberger & Welch, 1981). We drew inferences from the posterior mode and 95% Credible Intervals (95% CI).

Results

Temporal trend in arrival date

Between 1994 and 2020, the annual mean arrival date advanced at a rate of 0.36 days per year (overall change: -9.34 days, 95% CI: -10.43, -8.24; Figure 1A), as did the individuals' mean arrival date regressed over their mean breeding year, with a rate of 0.09 days per year (overall change: -2.45 days, 95% CI: -4.51, -0.38).

Sources of phenotypic variation in arrival date and adult LRS

The mean population age increased from 2 to 8 years over the first 15 years of the study, before leveling off (Supplementary Figure S2). Arrival date was influenced by the linear and quadratic effects of age, as well as by sea surface temperature at the main wintering area and year of breeding in interaction with age. On average, birds arrived earlier following years of warmer temperatures at the wintering area and in the later years of the study period, although these effects depended on the age of the individuals, with mature adult birds arriving earlier and showing a weaker age-dependent decline than younger birds (Table 1). Arrival date was also influenced by the sex of individuals: females arrived earlier than males (Table 1).

Interpreting the variance components, common terms showed strong heritable differences in arrival date: additive genetic effects accounted for 25.74% of the total phenotypic variance, while permanent environmental effects accounted for an additional 7.17% (Table 1). Arrival date also harbored variance among years of breeding (0.84%, Table 1). Maternal environmental and cohort effects were close to, or not significantly different from, zero (Table 1), although we may have limited statistical power to estimate the maternal environmental effect. Additionally, we found the additive genetic variance in adult LRS to be close to, or not significantly different from, zero (Table 1). While this is a pattern commonly observed in fitness and fitness components (reviewed by Hendry et al., 2018), the close-to-zero estimate might be due to either a true absence of additive genetic variance in fitness or a lack of power to detect it with higher precision. The latter case is true for several fitness components in this population (Moiron et al., 2022).

Estimating micro-evolutionary change and genetic drift

The best linear unbiased predictors (BLUPs) for breeding values of arrival date extracted from the bivariate animal model described above were tested for a trend over time, that is, over an individual's mean breeding year or across cohorts. Using cohort for the temporal trend, the slope of the linear regression was estimated at –0.021 days per year of breeding (95% CI: –0.056, 0.016), indicating that breeding values advanced ~0.4 days across the 27-year study (Figure 1B). While the lower 95% CI of the annual estimate was overlapping zero, the fraction of the posterior distribution of the genetic slopes that was greater than zero was only 14%.



Figure 1. Temporal phenotypic trends in mean arrival date across years (A) and in individual breeding values in arrival date across cohorts (i.e., year of hatching) (B). Breeding values were extracted from the bivariate animal model of arrival date and adult lifetime reproductive success reported in Table 1. In (A), large silver dots represent annual means, small grey dots individual observations, and silver lines the slope and associated 95% confidence interval of the phenotypic trend across the 27 years of study. In (B), grey dots represent individual breeding values of arrival date; black solid line represents no change in breeding values; silver lines are the slope and associated 95% confidence interval of the temporal trend in breeding values.

Table 1. Estimates from a bivariate animal model of arrival date and adult lifetime reproductive success (LRS	(S)
---	-----

Fixed effects	β (95% CI)	
Intercept _{arrival date}	156.356	[154.7, 158.439]
Intercept	0.857	[0.573, 1.191]
Age [linear]	-8.4	[-8.655, -8.058]
Age [quadratic]	0.368	[0.35, 0.385]
Sea surface temperature	-1.565	[-2.834, 0.265]
Sea surface temperature × age [linear]	0.158	[-0.055, 0.4]
Sea surface temperature × age [quadratic]	-0.013	[-0.023, -0.001]
Year	-4.302	[-5.669, -2.394]
Year × age [linear]	1.26	[1.071, 1.596]
Year × age [quadratic]	-0.102	[-0.12, -0.092]
Sex [female] arrival date	-1.637	[-2.221, -1.012]
Sex [female] fitness	0.1	[-0.066, 0.208]
Random effects	σ² (95% CI)	
	22 425	[15.343, 29.651]
V. c.	0.002	[0, 0, 268]
COV.	-0.057	[-1.055, 0.271]
V. comments and the state of th	0.011	[0, 2,334]
V. NOTHER ATTIVAL date	0.001	[0, 0, 123]
COV MOTURE arrival data Starsa	0	[-0.176, 0.115]
V COHORT arrival data	0.813	[0, 2.186]
	0.389	[0.166, 0.843]
COV COLORT arrival data fitness	-0.181	[-0.825, 0.178]
V _{PF arrival date}	8.807	[1.419, 12.828]
V _{YEAR arrival date}	7.135	[4.01, 14.715]
V _{R arrival date}	52.614	[50.918, 54.462]
V _{R fitness}	0.809	[0.621, 1.009]
COV _{PE arrival date - R fitness}	-0.636	[-1.38, -0.047]

Note. Estimates shown represent V_A for additive genetic variance, COV_A for additive genetic covariance, V_{PE} for permanent environmental variance, $(CO)V_{MOTPHER}$ for maternal environmental (co)variances, $(CO)V_{COHORT}$ for cohort (co)variances, V_{VEAR} for among-year variance, and V_g for residual variance. All continuous fixed effects, except age, were mean centered and variance standardized. Estimates represent posterior modes with associated 95% credible intervals.

Re-expressed in units of phenotypic standard deviations per generation, the estimated rate of micro-evolution corresponded to an observed evolutionary change of -0.003 Haldanes (95% CI: -0.012, 0.003), which can also be expressed as -0.129 days per generation (95% CI: -0.449, 0.126). Using an individual's mean breeding year for the temporal trend, we observed a similar pattern: the slope of

the linear regression was estimated at -0.021 days per year (95% CI: -0.074, 0.018, Supplementary Figure S4). Finally, less than 8.3% of the simulations of random drift generated an advance as large as, or larger than, the change estimated from the linear regression of genetic change using either cohort or an individual's mean breeding years (Supplementary Figure S3).

Predicting micro-evolutionary change

The phenotypic selection differential for arrival date quantified as the sum of additive genetic and permanent environmental covariances between the trait and fitness (Table 1) was negative and the associated 95% credible intervals did not overlap zero (S = -1.058, 95% CI: -1.609, -0.592), indicating that individuals with earlier arrival dates obtained, on average, higher adult LRS.

Given that arrival date was heritable and under directional selection (Table 1), the Breeder's equation predicted a rate of evolutionary change in arrival date of -0.029 days per year (95% CI: -0.050, -0.015), equivalent to a total change of -0.742 days across the 27 years of study (95% CI: -1.297, -0.432, Figure 2), and translating into a response rate of -0.228 days per generation (95% CI: -0.339, -0.119) and -0.018 Haldanes (95% CI: -0.032, -0.009).

The bivariate animal model of adult LRS and arrival date revealed a negative additive genetic covariance between arrival date and LRS ($COV_A = -0.057$, 95% CI: -1.055, 0.271, Table 1), corresponding to a total evolutionary change in arrival date of -0.185 days across the 27-year study period (95% CI: -3.430, 0.880; Figure 2), or a predicted evolutionary rate of -0.007 days per year (95% CI: -0.132, 0.034) and

-0.001 Haldanes (95% CI: -0.084, 0.021). While the point estimate for the Robertson's Secondary Theorem of natural Selection also predicted an advance in arrival date, the 95% CI of the predicted evolutionary response was wide and the lower 95% CI limit overlapped with zero (Table 1). Nevertheless, 82.2% of the posterior samples were below zero, suggesting the additive genetic covariance between arrival date and LRS to be very likely negative, but we might lack the statistical power to detect it with more certainty.

Discussion

In this study, we investigated a case of expected evolutionary change in the timing of spring migration of a seabird to further our understanding of the evolutionary dynamics of phenology in natural populations, whilst also unraveling the selection pressures that underlie micro-evolutionary changes. We did so by using long-term phenotypic data from a free-living population of common terns located at the North Sea coast of Germany and applying a bivariate animal model that allowed incorporating relatedness information from a social pedigree.



Figure 2. Posterior distributions for the estimated and predicted evolutionary response of arrival date over the 27-year study period. From top to bottom, (A) "Estimated genetic change" was measured as the temporal change in individual breeding values for arrival date, (B) the predicted rate of evolutionary response based on the "Breeder's equation" was quantified using a bivariate animal model from which we extracted the estimates of selection differential (individual-level covariance between arrival date and adult lifetime reproductive success) and heritability of arrival date, and (C) the predicted rate of evolutionary response based on the "Robertson's Secondary Theorem of Selection" was quantified as the additive genetic covariance between arrival date and adult lifetime reproductive success. All estimates presented were converted to an evolutionary change ("evolution") measured in days of change over the 27-year study period. Parameter estimates are summarized in the main text. All distributions have the same area and axes scales.

Previous studies from the common tern population at the Banter See found that the timing of migration from the wintering grounds was heritable ($h^2 = 0.06 \pm 0.03$, Arnaud et al., 2013), and under directional selection (Arnaud et al., 2013, but Ezard et al., 2007). In the current study, we corroborated those earlier findings using a considerably longer and larger dataset, finding a heritability of 24.7%, and a phenotypic selection differential of -1.1 days. Given that the prerequisites for a potential evolutionary change were met, we expected an evolutionary response to selection with advanced arrival dates from the wintering grounds, that is, a change in mean phenotype towards earlier arrival dates at the genetic level, although apparent evolutionary stasis is widespread (see Table 1 in Merilä et al., 2001 and Table 20.3 in Walsh & Lynch, 2018), and phenotypic plasticity and/or other sources of non-genetic variation might also lead to phenotypic divergence (e.g., Bonnet et al., 2017, 2019; Gienapp & Merilä, 2014; Gienapp et al., 2008). Indeed, a previous study from this population of common terns found support for individual, and, to a lesser extent, additive genetic variance in the plasticity of arrival date in response to an important climatic factor at the main wintering areas (Moiron et al. in revision), although that was not the case for the plasticity of timing of breeding (Dobson et al., 2017).

We found the average arrival date to have advanced 9.34 days across the 1994-2020 study period (95% CI ranged from -10.43 to -8.24, Figure 1A). Likewise, there was a temporal trend in the average breeding value for arrival date (Figure 1B). Taking into account the parameters' uncertainties and early criticisms on the use of predicted breeding values (Hadfield et al., 2010; Postma, 2006), we found that the expected response to selection ranged from -0.056 to 0.016 days per year when using cohort to assess the temporal trend. Whereas the 95% credible intervals of the observed genetic change overlapped with zero, the posterior probability of this change being greater than zero was only 14% (Figure 2). Additionally, the probability of random genetic drift as the sole driver of the evolutionary change to generate a trend in breeding values as strong as, or stronger than, the observed genetic trend was only 8.3%, indicating that an evolutionary response to selection is a much more likely explanation for the observed temporal change in arrival date than genetic drift.

Theoretical predictions of evolutionary change

We estimated all parameters associated with the Breeder's equation and Robertson's Secondary Theorem of Selection (STS), for example, heritability, selection differential, and genetic covariance between the focal phenotype and relative fitness, using a single Bayesian animal model. This approach allows for a quantitative comparison of both evolutionary change predictions, while also taking forward the uncertainties in the estimated parameters and subsequent calculations. The predicted rate of evolutionary change in arrival date based on the Breeder's equation was -0.018 Haldanes (95% CI: -0.032, -0.009), indicating that common terns were expected to advance the timing of their spring migration over time. The predicted rate of evolutionary change in arrival date based on the STS was -0.001 Haldanes, similarly indicating a response to selection towards earlier spring migration dates, although the associated 95% credible intervals for this estimate were overlapping zero (95% CI: -0.084, 0.021).

Altogether, the theoretical predictions of evolutionary change based on the Breeder's equation and Robertson's Secondary Theorem of Selection were largely concordant, both in direction and strength, although the STS estimate was associated with substantial statistical uncertainty. In addition, these two theoretical predictions of evolutionary change were qualitatively similar to the estimated genetic change (i.e., the posterior distributions for the three parameters largely overlapped, Figure 2), and in line with the observed temporal trend in the mean phenotype (Figure 1A). As such, our results provide a rare case of agreement in theoretical predictions of evolutionary change that are consistent with observed genetic patterns.

Additionally, and given that we found the evolutionary predictions from the Breeder's equation and STS to be in substantial agreement, our finding implies that the Breeder's equation might be a good predictor of evolutionary change in our study system, where the assumption of causality most likely holds, at least, when assuming a static environment (Kruuk et al., 2002, 2003; Morrissey et al., 2010, 2012a; Queller, 1992; Rausher, 1992). It also implies that our selection estimate might be mostly unbiased, that is, the individual-level covariance between arrival date and relative fitness is mostly caused by arrival date, and there are no "missing traits" (Hadfield, 2008; Morrissey et al., 2010, 2012b; Queller, 1992). These results shed light on the true form of natural selection acting on spring migration timing and suggest a lack of genetic constraints that might interfere with an evolutionary response to selection. However, it is important to note that, while our three estimates of evolutionary response largely agreed, conclusions must be drawn with care as we cannot readily neglect the potential for the apparent agreement in evolutionary predictions to be coincidental. Multiple factors can potentially bias evolution estimates (e.g., a "missing fraction" in our fitness estimate or indirect selection from genetically-correlated traits), most often affecting evolutionary predictions based on the Breeder's equation, but also on STS (Kruuk et al., 2002, 2003; Merilä et al., 2001; Morrissey et al., 2012a; Walsh & Lynch, 2018).

Global warming as underlying driver of phenotypic change

Temperatures in the wintering grounds of the studied population have significantly increased since 1994 (sea surface temperature change in West Africa between 1994 and 2020 = 0.66°C, 95% CI: 0.16, 1.16), and such warming has been associated with earlier spring phenological events (Dobson et al., 2017), making climate warming an obvious mechanism potentially underlying the population's change in migration timing. However, given the observational nature of our study and the complex life cycle of long-distance migratory species experiencing environmental conditions in both hemispheres, we cannot test this hypothesis directly and identify the true biological cue. We can, however, speculate about the potential benefits of earlier arrival from the wintering grounds. Such early arrival could facilitate access to higher-quality breeding sites (Kokko et al., 2004) and mates (Coppack & Both, 2002; Ludwigs & Becker, 2005), and an early initiation of breeding. Early breeding, in turn, would allow for the production of replacement clutches in cases of predation, accidental egg or chick loss, or poor food availability for rearing offspring early in the season, and such replacement clutches (8.9% of clutches across the study period) indeed are known to significantly contribute to LRS (Becker & Zhang, 2011). As such, selection for earlier timing of breeding may translate into selection for earlier timing of spring migration. Additionally, evolutionary responses in arrival date likely are aided by assortative mating for migration timing (Bearhop et al., 2005; see also Moiron et al., 2020).

Relative contributions of evolutionary and plastic responses to phenotypic change

The estimated genetic change in the timing of spring migration represented c. 5% of the population's phenotypic change, indicating that the observed advance in migration timing was mostly underpinned by other sources of environmental (non-genetic) variation, such as changes in the population's age structure (Supplementary Figure S2) or phenotypic plasticity. Indeed, a recent study identified a plastic response of arrival date to climate change, with birds arriving earlier at the breeding ground following warmer years at the main wintering area (Moiron et al., in revision). Altogether, our findings therefore are fully in line with general reports of phenotypic plasticity underlying the vast majority of phenotypic responses to climate change (Charmantier & Gienapp, 2014; Merilä & Hendry, 2014; Parmesan, 2006 and references therein; Vedder et al., 2013), but also highlight that the existence of phenotypic plasticity does not readily imply a lack of a micro-evolutionary response or preclude their simultaneous existence, even in relatively long-lived species (e.g., Bonnet et al., 2019).

Concluding remarks

While plastic responses to climate change are widely reported across taxonomic groups, evidence of micro-evolutionary responses to selection is still uncommon, and considerably more challenging to accrue, mostly because the data collection for documenting micro-evolution in the wild remains a longterm task. In this study, we provided rare evidence of an evolutionary change in spring phenology, likely in response to global warming, while also illustrating how both plastic and genetic changes can simultaneously facilitate phenotypic divergence in natural populations. Our work further emphasizes the importance of maintaining long-term individual-based studies of natural populations to unravel the diversity of biological responses to climate change, and, generally, to understand selective patterns and evolutionary dynamics of phenotypic traits in the wild.

Supplementary material

Supplementary material is available online at *Evolution Letters* (https://academic.oup.com/evlett/qrad014).

Data availability

Data has been archived in the Dryad Digital Repository (https://doi.org/10.5061/dryad.8kprr4xqj).

Author contributions

M.M. conceived the study with input from S.B., A.C. and C.T. M.M. designed and conducted the statistical analyses, and wrote the manuscript. S.B. manages the tern data and collated the phenotypic dataset. B.H. contributed the sea surface temperature dataset. All authors contributed to discussions and editing of the final paper.

Funding

M.M. was funded by a Marie Skłodowska Curie Individual Fellowship (PLASTIC TERN, Grant Agreement Number 793550) and an Alexander von Humboldt Research Fellowship for Postdoctoral Researchers.

Conflict of interest: The authors declare no conflict of interest. Editorial processing of the manuscript was done independently of C.T. and A.C. who are associate editors of Evolution Letters.

Acknowledgments

We are indebted to Peter H. Becker for setting up the long-term common tern study and thank the numerous students and field workers that have contributed to compiling the data. We also thank Timothée Bonnet and an anonymous reviewer for their constructive feedback on the manuscript.

References

- Åkesson, S., Ilieva, M., Karagicheva, J., Rakhimberdiev, E., Tomotani, B., & Helm, B. (2017). Timing avian long-distance migration: From internal clock mechanisms to global flights. *Philosophical Transactions of the Royal Society B Biological Science*, **372**, 20160252.
- Arnaud, C. M., Becker, P. H., Dobson, F. S., & Charmantier, A. (2013). Canalization of phenology in common terns: Genetic and phenotypic variations in spring arrival date. *Behavioral Ecology*, 24, 683–690.
- Bailey, L. D., van de Pol, M.M., Adriaensen, F., Arct, A., Barba, E., Bellamy, P. E., Bonamour, S., Bouvier, J.-C., Burgess, M. D., Charmantier, A., Cusimano, C., Doligez, B., Drobniak, S. M., Dubiec, A., Eens, M., Eeva, T., Ferns, P. N., Goodenough, A. E., Hartley, I. R., Hinsley, S. A., ... Visser, M. E. (2022). Bird populations most exposed to climate change are less sensitive to climatic variation. Nature Communications, 13, 2112.
- Bearhop, S., Fiedler, W., Furness, R. W., Votier, S. C., Waldron, S., Newton, J., Bowen, G. J., Berthold, P., & Farnsworth, K. (2005). Assortative mating as a mechanism for rapid evolution of a migratory divide. *Science*, **310**, 502–504.
- Becker, P. H., Schmaljohann, H., Riechert, J., Wagenknecht, G., Zajková, Z., & González-Solís, J. (2016). Common terns on the east Atlantic flyway: Temporal–spatial distribution during the non-breeding period. *Journal fuer Ornithologie*, **157**(4), 927–940. https://doi.org/10.1007/s10336-016-1346-2
- Becker, P. H., & Wendeln, H. (1997). A new application for transponders in population ecology of the common tern. Condor, 99, 534–538.
- Becker, P. H., & Wink, M. (2003). Influences of sex, sex composition of brood and hatching order on mass growth in common terns Sterna Hirundo. Behavior, Ecology and Sociobiology, 54(2), 136–146. https://doi.org/10.1007/s00265-003-0605-4
- Becker, P. H., & Zhang, H. (2011). Renesting of common terms sterna Hirundo in the life history perspective. *Journal fuer Ornithologie*, 152, 213–225.
- Bellard, C., Bertelsmeier, C., Leadley, P., Thuiller, W., & Courchamp, F. (2012). Impacts of climate change on the future of biodiversity. Ecology Letters, 15(4), 365–377. https://doi. org/10.1111/j.1461-0248.2011.01736.x
- Bonnet, T., Morrissey, M. B., Morris, A., Morris, S., Clutton-Brock, T. H., Pemberton, J. M., & Kruuk, L. E. B. (2019). The role of selection and evolution in changing parturition date in a red deer population. PLoS Biology, 17, e30004931–e30004923. https://doi. org/10.1371/journal.pbio.3000493
- Bonnet, T., Wandeler, P., Camenisch, G., & Postma, E. (2017). Bigger is fitter? Quantitative genetic decomposition of selection reveals an adaptive evolutionary decline of body mass in a wild rodent population. PLoS Biology, 15, e10025921–e10025921. https://doi. org/10.1371/journal.pbio.1002592
- Bouwhuis, S., Vedder, O., & Becker, P. H. (2015). Sex-specific pathways of parental age effects on offspring lifetime reproductive success in a long-lived seabird. *Evolution*, **69**, 1760–1771.
- Charlesworth, B. (1994). Evolution in age-structured populations. Cambridge University Press.

- Charmantier, A., & Gienapp, P. (2014). Climate change and timing of avian breeding and migration: Evolutionary versus plastic changes. *Evolutionary Applications*, **7**(1), 15–28. https://doi. org/10.1111/eva.12126
- Chen, I. -C., Hill, J. K., Ohlemüller, R., Roy, D. B., & Thomas, C. D. (2011). Rapid range shifts of species associated with high levels of climate warming. *Science*, **333**, 1024–1026.
- Coppack, T., & Both, C. (2002). Predicting life-cycle adaptation of migratory birds to global climate change. *Ardea*, **90**, 369–378.
- Dobson, F. S., Becker, P. H., Arnaud, C. M., Bouwhuis, S., & Charmantier, A. (2017). Plasticity results in delayed breeding in a long-distant migrant seabird. Ecology and Evolution, 7(9), 3100–3109. https:// doi.org/10.1002/ece3.2777
- Evans, S. R., & Gustafsson, L. (2017). Climate change upends selection on ornamentation in a wild bird. *Nature Ecology and Evolution*, **1**, 0039.
- Ezard, T. H. G., Becker, P. H., & Coulson, T. (2007). Correlations between age, phenotype, and individual contribution to population growth in common terns. *Ecology*, 88(10), 2496–2504. https://doi.org/10.1890/06-2020.1
- Falconer, D. S., & Mackay, T. F. C. (1996). Introduction to quantitative genetics (4th ed.). London, Prentice Hall.
- Gienapp, P., & Brommer, J. E. (2014). Evolutionary dynamics in response to climate change. In A. Charmantier, D. Garant, & L.
 E. B. Kruuk (Eds.), *Quantitative genetics in the wild* (pp. 254–274).
 [Chap. 15] Oxford University Press.
- Gienapp, P., Leimu, R., & Merilä, J. (2007). Responses to climate change in avian migration time: Microevolution versus phenotypic plasticity. Climate Research, 35, 25–35. https://doi.org/10.3354/cr00712
- Gienapp, P., & Merilä, J. (2014). Disentangling plastic and genetic changes in body mass of Siberian jays. *Journal of Evolutionary Biology*, 27(9), 1849–1858. https://doi.org/10.1111/jeb.12438
- Gienapp, P., Reed, T. E., & Visser, M. E. (2014). Why climate change will invariably alter selection pressures on phenology. *Proceedings of the Royal Society B: Biological Sciences*, **281**(1793), 20141611.
- Gienapp, P., Teplitsky, C., Alho, J. S., Mills, J. A., & Merilä, J. (2008). Climate change and evolution: Disentangling environmental and genetic responses. *Molecular Ecology*, **17**(1), 167–178. https:// doi.org/10.1111/j.1365-294X.2007.03413.x
- Gordo, O. (2007). Why are bird migration dates shifting? A review of weather and climate effects on avian migratory phenology. *Climate Research*, **35**, 37–58. https://doi.org/10.3354/cr00713
- Hadfield, J. D. (2008). Estimating evolutionary parameters when viability selection is operating. *Proceedings of the Royal Society B:* Biological Sciences, **275**, 723–734.
- Hadfield, J. D. (2010). MCMC methods for multi-response generalized linear mixed models: The MCMCglmm R package. *Journal of the Statistical Software*, **33**, 1–22.
- Hadfield, J. D., Wilson, A. J., Garant, D., Sheldon, B. C., & Kruuk, L. E. B. (2010). The misuse of BLUP in ecology and evolution. American Naturalist, 175, 116–125.
- Hairston, N. G., Ellner, S. P., Geber, M. A., Yoshida, T., & Fox, J. A. (2005). Rapid evolution and the convergence of ecological and evolutionary time. *Ecology Letter*, 8, 1114–1127.
- Haest, B., Hüppop, O., & Bairlein, F. (2018). The influence of weather on avian spring migration phenology: What, where and when? *Global Change Biology*, 24, 5769–5788.
- Haest, B., Hüppop, O., & Bairlein, F. (2020). Weather at the winter and stopover areas determines spring migration onset, progress, and advancements in Afro-Palearctic migrant birds. Proceedings of the National Academy of Sciences, **117**, 17056–17062.
- Haest, B., Hüppop, O., Pol, M., & Bairlein, F. (2019). Autumn bird migration phenology: A potpourri of wind, precipitation and temperature effects. *Global Change Biology*, **25**, 4064–4080.

- Heidelberger, P., & Welch, P. D. (1981). A spectral method for confidence interval generation and run length control in simulations. Communications of the ACM, 24(4), 233–245. https://doi. org/10.1145/358598.358630
- Helm, B., Van Doren, B. M., Hoffmann, D., & Hoffmann, U. (2019). Evolutionary response to climate change in migratory pied flycatchers. *Current Biology*, **29**(21), 3714–3719.
- Henderson, C. R. (1963). Selection index and expected genetic advance. Statistics and Plant Breeding, 141–163.
- Hendry, A. P., Farrugia, T. J., & Kinnison, M. T. (2008). Human influences on rates of phenotypic change in wild animal populations. *Molecular Ecology*, **17**(1), 20–29. https://doi. org/10.1111/j.1365-294X.2007.03428.x
- Hendry, A. P., & Kinnison, M. T. (1999). Perspective: The pace of modern life: Measuring rates of contemporary microevolution. *Evolution*, **53**, 1637–1653.
- Hendry, A. P., Schoen, D. J., Wolak, M. E., & Reid, J. M. (2018). The contemporary evolution of fitness. Annual Review of Ecology, Evolution, and Systematics, 49(1), 457–476. https://doi.org/10.1146/ annurev-ecolsys-110617-062358
- Hoffmann, A. A., & Sgrò, C. M. (2011). Climate change and evolutionary adaptation. Nature, **470**(7335), 479–485. https://doi. org/10.1038/nature09670
- Keogan, K., Daunt, F., Wanless, S., Phillips, R. A., Walling, C. A., Agnew, P., Ainley, D. G., Anker-Nilssen, T., Ballard, G., Barrett, R. T., Barton, K. J., Bech, C., Becker, P., Berglund, P. A., Bollache, L., Bond, A. L., Bouwhuis, S., Bradley, R. W., Burr, Z. M., ... Lewis, S. (2018). Global phenological insensitivity to shifting ocean temperatures among seabirds. *Nature Climate Change*, 8, 313–317. Springer US.
- Kokko, H., Harris, M. P., & Wanless, S. (2004). Competition for breeding sites and site-dependent population regulation in a highly colonial seabird, the common guillemot Uria aalge. *Journal of Animal Ecology*, **73**(2), 367–376. https://doi. org/10.1111/j.0021-8790.2004.00813.x
- Kruuk, L. E. B. (2004). Estimating genetic parameters in natural populations using the "animal model". Philosophical Transactions of the Royal Society B Biological Science, **359**, 873–890.
- Kruuk, L. E. B., Merilä, J., & Sheldon, B. C. (2003). When environmental variation short-circuits natural selection. *Trends in Ecology and Evolution*, **18**, 207–209.
- Kruuk, L. E. B., Slate, J., Pemberton, J. M., Brotherstone, S., Guinness, F., & Clutton-Brock, T. (2002). Antler size in red deer: heritability and selection but no evolution. *Evolution*, **56**, 1683–1695.
- Kürten, N., Schmaljohann, H., Bichet, C., Haest, B., Vedder, O., González-Solís, J., & Bouwhuis, S. (2022). High individual repeatability of the migratory behaviour of a long-distance migratory seabird. *Movement Ecology*, **10**, 5.
- Lavergne, S., Mouquet, N., Thuiller, W., & Ronce, O. (2010). Biodiversity and climate change: Integrating evolutionary and ecological responses of species and communities. *Annual Review of Ecology, Evolution, and Systematics*, **41**(1), 321–350. https://doi.org/10.1146/ annurev-ecolsys-102209-144628
- Lehikoinen, E., Sparks, T. H., & Zalakevicius, M. (2004). Arrival and departure dates. Advances in Ecological Research **35**, 1–31.
- Ludwigs, J. -D., & Becker, P. H. (2005). What do pairing patterns in common tern, Sterna hirundo, recruits reveal about the significance of sex and breeding experience? Behavior, Ecology and Sociobiology, 57, 412–421.
- Lynch, M., & Walsh, B. (1998). Genetics and analysis of quantitative trait. Sinauer Associates.
- Menzel, A., Sparks, T. H., Estrella, N., Koch, E., Aasa, A., Ahas, R., Alm-Kübler, K., Bissolli, P., Braslavská, O., Briede, A., Chmielewski,

F. M., Crepinsek, Z., Curnel, Y., Dahl, A., Defila, C., Donnelly, A., Filella, Y., Jatczak, K., Måge, F., ... Zust, A. (2006). European phenological response to climate change matches the warming pattern. *Global Change Biology*, **12**, 1969–1976.

- Merilä, J., & Hendry, A. P. (2014). Climate change, adaptation, and phenotypic plasticity: The problem and the evidence. *Evolutionary Applications*, **7**(1), 1–14. https://doi.org/10.1111/eva.12137
- Merilä, J., Sheldon, B. C., & Kruuk, L. E. B. (2001). Explaining stasis: Microevolutionary studies in natural populations. *Genetica*, **112**, 199–222.
- Moiron, M., Araya-Ajoy, Y. G., Teplitsky, C., Bouwhuis, S., & Charmantier, A. (2020). Understanding the social dynamics of breeding phenology: Indirect genetic effects and assortative mating in a long-distance migrant. *The American Naturalist*, **196**(5), 566–576.
- Moiron, M., Charmantier, A., & Bouwhuis, S. (2022). The quantitative genetics of fitness in a wild seabird. *Evolution*, **76**, 1443–1452.
- Morrissey, M. B., & Bonnet, T. (2019). Analogues of the fundamental and secondary theorems of selection, assuming a log-normal distribution of expected fitness. *Journal of Heredity*, **110**(4), 396– 402. https://doi.org/10.1093/jhered/esz020
- Morrissey, M. B., Kruuk, L. E. B., & Wilson, A. J. (2010). The danger of applying the breeder's equation in observational studies of natural populations. *Journal of Evolutionary Biology*, **23**(11), 2277– 2288. https://doi.org/10.1111/j.1420-9101.2010.02084.x
- Morrissey, M. B., Parker, D. J., Korsten, P., Pemberton, J. M., Kruuk, L. E., & Wilson, A. J. (2012a). The prediction of adaptive evolution: Empirical application of the secondary theorem of selection and comparison to the breeder's equation. *Evolution*, 66(8), 2399–2410.
- Morrissey, M. B., Walling, C. A., Wilson, A. J., Pemberton, J. M., Clutton-Brock, T. H., & Kruuk, L. E. B. (2012b). Genetic analysis of life-history constraint and evolution in a wild ungulate population. American Naturalist, **179**(4), E97–E114. https://doi. org/10.1086/664686
- Nisbet, I. C. T., Arnold, J. M., Oswald, S. A., Pyle, P., & Patten, M. A. (2020). Common tern (Sterna hirundo). In S. M. Billerman (Ed.), Birds of the world. Cornell Lab of Ornithology.
- Parmesan, C. (2006). Ecological and evolutionary responses to recent climate change. Annual Review of Ecology, Evolution, and Systematics, **37**(1), 637–669. https://doi.org/10.1146/annurev. ecolsys.37.091305.110100
- Parmesan, C., & Yohe, G. (2003). A globally coherent fingerprint of climate change impacts across natural systems. Nature, 421(6918), 37–42. https://doi.org/10.1038/nature01286
- Poloczanska, E. S., Brown, C. J., Sydeman, W. J., Kiessling, W., Schoeman, D. S., Moore, P. J., Brander, K., Bruno, J. F., Buckley, L. B., Burrows, M. T., Duarte, C. M., Halpern, B. S., Holding, J., Kappel, C. V., O'Connor, M. I., Pandolfi, J. M., Parmesan, C., Schwing, F., Thompson, S. A., & Richardson, A. J. (2013). Global imprint of climate change on marine life. Nature Climate Change, 3, 919–925.
- Postma, E. (2006). Implications of the difference between true and predicted breeding values for the study of natural selection and micro-evolution. *Journal of Evolutionary Biology*, **19**(2), 309–320. https://doi.org/10.1111/j.1420-9101.2005.01007.x
- Price, G. R. (1970). Selection and covariance. *Nature*, **227**(5257), 520–521. https://doi.org/10.1038/227520a0
- Pulido, F. (2007). Phenotypic changes in spring arrival: Evolution, phenotypic plasticity, effects of weather and condition. Climate Research, 35, 5–23. https://doi.org/10.3354/cr00711
- Queller, D. C. (1992). Quantitative genetics, inclusive fitness, and group selection. American Naturalist, 139(3), 540–558. https://doi. org/10.1086/285343

- R Core Team. (2019). A language and environment for statistical computing. R Foundation for Statistical Computing.
- Radchuk, V., Reed, T., Teplitsky, C., van de Pol, M., Charmantier, A., Hassall, C., Adamík, P., Adriaensen, F., Ahola, M. P., Arcese, P., Miguel Avilés, J., Balbontin, J., Berg, K. S., Borras, A., Burthe, S., Clobert, J., Dehnhard, N., de Lope, F., Dhondt, A. A., ... Kramer-Schadt, S. (2019). Adaptive responses of animals to climate change are most likely insufficient. Nature Communications, **10**, 1–14.
- Rausher, M. D. (1992). The measurement of selection on quantitative traits: Biases due to environmental covariances between traits and fitness. *Evolution*, **46**, 616–626.
- Robertson, A. (1966). A mathematical model of the culling process in dairy cattle. *Animal Science*, **8**(1), 95–108. https://doi.org/10.1017/s0003356100037752
- Root, T. L., Price, J. T., Hall, K. R., Schneider, S. H., Rosenzweig, C., & Pounds, J. A. (2003). Fingerprints of global warming on wild animals and plants. *Nature*, **421**(6918), 57–60. https://doi. org/10.1038/nature01333
- Salamin, N., Wüest, R. O., Lavergne, S., Thuiller, W., & Pearman, P. B. (2010). Assessing rapid evolution in a changing environment. Trends in Ecology and Evolution, 25(12), 692–698. https://doi. org/10.1016/j.tree.2010.09.009
- Szostek, K. L., & Becker, P. H. (2012). Terns in trouble: Demographic consequences of low breeding success and recruitment on a common tern population in the German Wadden Sea. *Journal fuer Ornithologie*, **153**, 313–326.
- Tarka, M., Hansson, B., & Hasselquist, D. (2015). Selection and evolutionary potential of spring arrival phenology in males and females of a migratory songbird. *Journal of Evolutionary Biology*, 28(5), 1024–1038. https://doi.org/10.1111/jeb.12638
- Teplitsky, C., Mouawad, N. G., Balbontin, J., De Lope, F., & Møller, A. P. (2011). Quantitative genetics of migration syndromes: A study of two barn swallow populations. *Journal of Evolutionary Biology*, 24(9), 2025–2039. https://doi.org/10.1111/j.1420-9101.2011.02342.x
- Thackeray, S. J., Sparks, T. H., Frederiksen, M., Burthe, S., Bacon, P. J., Bell, J. R., Botham, M. S., Brereton, T. M., Bright, P. W., Carvalho, L., Clutton-Brock, T., Dawson, A., Edwards, M., Elliott, J. M., Harrington, R., Johns, D., Jones, I. D., Jones, J. T., Leech, D. I., ... Wanless, S. (2010). Trophic level asynchrony in rates of phenological change for marine, freshwater and terrestrial environments. Global Change Biology, 16, 3304–3313.
- Vedder, O., Bouwhuis, S., & Sheldon, B. C. (2013). Quantitative assessment of the importance of phenotypic plasticity in adaptation to climate change in wild bird populations. PLoS Biology, 11, e10016051–e10016010. https://doi.org/10.1371/journal. pbio.1001605
- de Villemereuil, P., Schielzeth, H., Nakagawa, S., & Morrissey, M. (2016). General methods for evolutionary quantitative genetic inference from generalized mixed models. *Genetics*, **204**(3), 1281–1294. https://doi.org/10.1534/genetics.115.186536
- Visser, M. E. (2008). Keeping up with a warming world; assessing the rate of adaptation to climate change. *Proceedings of the Royal Society B: Biological Sciences*, **275**, 649–659.
- Walsh, B., & Lynch, M. (2018). Evolution and selection of quantitative traits. Oxford University Press.
- Zhang, H., Vedder, O., Becker, P. H., & Bouwhuis, S. (2015a). Agedependent trait variation: The relative contribution of within-individual change, selective appearance and disappearance in a long-lived seabird. *Journal of Animal Ecology*, 84(3), 797–807. https://doi.org/10.1111/1365-2656.12321
- Zhang, H., Vedder, O., Becker, P. H., & Bouwhuis, S. (2015b). Contrasting between- and within-individual trait effects on mortality risk in a long-lived seabird. Ecology, 96(1), 71–79. https://doi. org/10.1890/14-0064.1